Author's response to reviews

Title: Assessment and management of neurogenic claudication in a specialised primary care musculoskeletal service: a survey of current practice among physiotherapists

Authors:

Christine M Comer (c.comer@nhs.net)
Anthony C Redmond (A.Redmond@leeds.ac.uk)
Howard A Bird (Howard.Bird@leedsth.nhs.uk)
Philip G Conaghan (p.conaghan@leeds.ac.uk)

Version: 3 Date: 9 June 2009

Author's response to reviews: see over
Dear Cintia

Re: Manuscript revision submission - Assessment and management of neurogenic claudication in a specialised primary care musculoskeletal service: a survey of current practice among physiotherapists - Christine Comer, Anthony Redmond, Howard Bird, Philip Conaghan

Please find enclosed our revised manuscript. Further to our e-mail correspondence, a revision has been made under the ‘procedures’ heading of the Methods section in response to your query regarding ethical approval and consent.

Please contact us should you have any further queries regarding this manuscript.

Yours sincerely

Christine Comer